

NOTA DI LAVORO 121.2010

Harmful Signaling in Matching Markets

By **Alexey Kushnir**, Economics Department, Pennsylvania State University

SUSTAINABLE DEVELOPMENT Series Editor: Carlo Carraro

Harmful Signaling in Matching Markets

By Alexey Kushnir, Economics Department, Pennsylvania State University

Summary

Some labor markets have recently developed formal signalling mechanisms, e.g. the signalling for interviews in the job market for new Ph.D. economists. We evaluate the effect of such mechanisms on two-sided matching markets by considering a game of incomplete information between firms and workers. Workers have almost aligned preferences over firms: each worker has "typical" commonly known preferences with probability close to one and "atypical" idiosyncratic preferences with the complementary probability close to zero. Firms have some commonly known preferences over workers. We show that the introduction of a signalling mechanism is harmful for this environment. Though signals transmit previously unavailable information, they also facilitate information asymmetry that leads to coordination failures. As a result, the introduction of a signalling mechanism lessens the expected number of matches when signals are informative.

Keywords: Signaling, Cheaptalk, Matching

JEL Classification: C72, C78, D80, J44

This paper has been presented at the 15th Coalition Theory Network Workshop held in Marseille, France, on June 17-18, 2010 and organised by the Groupement de Recherche en Economie Quantitative d'Aix-Marseille, (GREQAM) <u>http://www.feem-web.it/ctn/events/10_Marseilles/ctn15i.htm</u>).

I am especially grateful to Vijay Krishna and Marek Pycia for advising me during this project. This paper has benefited from suggestions made by Kalyan Chatterjee, Ed Green, Tymofiy Mylovanov, and Neil Wallace.

Address for correspondence:

Alexey Kushnir Economics Department Pennsylvania State University 306 Kern Building University Park, PA 16802 USA E-mail: aik116@psu.edu

Harmful Signaling in Matching Markets

Alexey Kushnir*

December 14, 2009

Abstract

Some labor markets have recently developed formal signaling mechanisms, e.g. the signaling for interviews in the job market for new Ph.D. economists. We evaluate the effect of such mechanisms on two-sided matching markets by considering a game of incomplete information between firms and workers. Workers have almost aligned preferences over firms: each worker has "typical" commonly known preferences with probability close to one and "atypical" idiosyncratic preferences with the complementary probability close to zero. Firms have some commonly known preferences over workers. We show that the introduction of a signaling mechanism is harmful for this environment. Though signals transmit previously unavailable information, they also facilitate information asymmetry that leads to coordination failures. As a result, the introduction of a signaling mechanism lessens the expected number of matches when signals are informative.

JEL classification: C72, C78, D80, J44.

Key words: signaling, cheaptalk, matching.

^{*}Economics Department, Pennsylvania State University, 306 Kern Building, University Park, PA 16802, (email: aik116@psu.edu). I am especially grateful to Vijay Krishna and Marek Pycia for advising me during this project. This paper has benefited from suggestions made by Kalyan Chatterjee, Ed Green, Tymofiy Mylovanov, and Neil Wallace.

1 Introduction

In December 2006, the Ad Hoc Committee of American Economic Association (AEA) implemented signaling as an actual instrument to facilitate match formation in the job market for new Ph.D. economists.¹ This market begins in early Fall each year, when economics departments advertise open faculty positions and graduate students nearing completion of their dissertations apply for these positions according to their preferences. Each student has an opportunity to send two signals to two departments prior to the market. Each signal states only that the student has indicated her interest to a given department. The signals are private, and only the faculty of the chosen department knows the student's signal. On a later stage, departments invite students to interviews and daylong university visits that followed by job offers to selected candidates. However, each department can interview only a small portion of available students, which creates congestion in the market and introduces a strategic dimension to the interviewing process. An average department probably does not want spend time interviewing candidates who are being interviewed by the elite departments.

The Ad Hoc Committee introduced the signals in order to alleviate congestion at the interviewing stage. Roth (2008) suggests that the limited number of signals can credibly transmit information about students' preference, which could help to reduce the coordination failures faced by the market participants and facilitate better match formation (see also "Signaling for Interviews in the Economics Job Market" AEA (2005)² for more discussion). Coles et al. (2009) also obtain results that support this intuition. Contrary to these studies, this paper shows that the signals can impede match formation in some environments.

We analyze one-to-one matching market between firms and workers without transfers. Each agent knows its own preferences over agents on the other side of the market, but is uncertain about the preferences of other agents. Workers have almost aligned preferences. Each worker has either "typical" commonly known preferences with a probability close to one or "atypical" preferences taken from some distribution with the complementary probability close to zero. The preferences of workers are ex-ante independently distributed.³ Firms have some fixed and commonly known preferences over workers.

We concentrate on the analysis of a preference signaling mechanism by assuming that each agent knows its preferences. Therefore, the interviewing stage is not necessary. We

¹The Ad Hoc Committee was established in 2005 in order to develop ways to facilitate the job market for new Ph.D. economists. Its members are Alvin E. Roth (chair), John Cawley, Philip Levine, Muriel Niederle, and John Siegfried. The signaling mechanism is implemented via the AEA website: http://www.aeaweb.org/joe/signal/.

²The document was created by the Ad Hoc Committee to provide advice to participants in the job market for new Ph.D. economists; http://www.aeaweb.org/joe/signal/signaling.pdf.

³We assume that typical workers rank firms according to some public ranking. For example, typical candidates in the job market for new Ph.D. economists rank departments of economics in their field according to the U.S. News and World Report ranking.

consider a decentralized matching game with three stages. First, each worker chooses a firm, to which she sends her signal. Each worker sends up to one signal; workers send signals simultaneously. Only firms that receive signals observe them. Second, firms make decisions about job offers by taking into account signals received at the first stage. Each firm can make only one offer. Finally, each worker chooses an offer to accept among the available offers. Each worker can accept at most one offer.

We show that if firms respond to signals in this environment, i.e. treat signals informatively, the introduction of signals decreases the expected number of matches. Prior to the signaling, all firms have almost identical beliefs about worker preferences. However, after the signals are received firms may have disparate beliefs. This disparity in beliefs leads to coordination failure. As a result, the introduction of a signaling mechanism may decrease the total number of matches and the welfare of agents.

Finally, we examine the implications from the introduction of a mechanism with public signals, i.e. signals observed by all firms. Though the expected number of matches increases compared to the game with private signals, public signals still impede match formation for some environments. Public signals do not transmit enough information about worker preferences. This induces some firms to compete for the same workers, which creates mismatches and decreases the expected total number of matches.

In addition to the market of new Ph.D. economists, some versions of the signaling mechanism studied in this paper emerged in other markets. Some online dating websites employ signaling mechanisms, wherein agents send signals to potential partners to indicate special interest. For example, each man of Cupid.com has several "virtual roses" they can attach to their messages. Since there is almost no cost associated with an additional message to a woman, these scarce roses provide men with a means of indicating their genuine interest to particular women. Informal signaling is also an important part of the market for clinical psychologists, described by Roth and Xing (1997). The ability of candidates to convey information about the likelihood to accept an offer is crucial in this market. Program directors for internships in clinical psychology have a tendency to hire applicants who explicitly express their readiness to accept an offer immediately, even if these applicants are of a low quality. Finally, early college admission in the U.S. can be viewed as a form of preference signaling. Many schools require that applicants not send early applications to other schools and view an early application as a signal of a student's enthusiasm for a particular school.

Signaling that we analyze in this paper is a form of costless communication, or cheap talk. There is no penalty attached for lying, and claims do not directly affect payoffs.⁴ Therefore, signals can only enlarge the set of equilibria. In contrast to this paper, Crawford and Sobel

 $^{{}^{4}}$ Ration talk is a better name for signals in our setting. Though signals are costless, an agent can send only a limited number of signals.

(1982) show that a cheap talk lead to new equilibria that are Pareto-superior to the one without communication.

Our negative results on expected number of matches in new cheap talk equilibria is in line with Farrell and Gibbons (1989)'s results, though it differs in its intuition.⁵ Costless communication in their two-agent bargaining model gives the buyer an opportunity to pretend to have a lower value and the seller an opportunity to pretend to have a higher value (compared to the truthful information transmission in our model). This enhances their bargaining positions at the cost of the risk of no trade. New cheap talk equilibria are characterized by both less trade and a reduction in the expected gains from trade.

In addition, signaling in our paper differs fundamentally from the "job market signaling" first introduced in Spence (1973). First, signals in our model have no direct cost. In addition, signals serve to convey information about worker preferences rather than worker quality.

Some recent works also analyze variants of signaling mechanisms. Coles et al. (2009) is a paper that is most related to ours. They consider the same three stage model of decentralized two-sided matching markets. However, they analyze markets with firm "segments". Workers agree on the ranking of firms across segments, but have idiosyncratic and uniformly distributed preferences within segments. For instance, all workers may agree as to which firms are in the "top five" segment and which are in the "six to ten" segment, etc., but may disagree as to the exact ranking within a segment. Firm preferences over workers are idiosyncratic and uniformly distributed. They show that, on average, introducing a signaling mechanism increases both the expected number of matches as well as the expected welfare of workers for this environment. The welfare of firms, on the other hand, changes ambiguously. Coles et al. (2009) also examine the impact of a signaling mechanism for different environments that vary in market size, the number of signals, positions, and periods of interaction.

Avery and Levin (2009) analyze match formation in the U.S. college admission market. There are two types of early admission programs: *early action* programs, where students may apply early but without any commitment to enroll, and *early decision* programs, where students commit to enroll if accepted. Avery and Levin (2009) analyze early applications as a signal of a student's enthusiasm for a particular school. At the same time, schools benefit more from accepting more enthusiastic students. Their results suggest that selective (or elite) schools benefit from adopting early action policy. At the same time, a lower ranked school, by adopting early decision policy, can attract some highly qualified but cautious students, drawing them away from highly ranked schools.

Lee et al. (2009) run a field experiment with the major Korean online dating website Couple.net to measure potential gains from introducing "virtual roses".⁶ They find that

⁵We are thankful for Lones Smith who drew our attention to this comparison.

⁶This "virtual roses" mechanism is similar to that of Cupid.com described above.

users of both genders are more likely to accept a dating request when a "virtual rose" is attached, and that roses are most helpful in improving acceptance rates for senders of "below average popularity."

Preference signaling can also be useful in centralized matching markets. Abdulkadiroglu et al. (2008) show that the introduction of a signaling technology can improve the ex-ante efficiency of the deferred acceptance algorithm (see Gale and Shapley, 1962) in case of weak preferences. Lee and Schwarz (2007) analyze preferences signaling in centralized matching markets in a three step matching formation process: preference signaling, investments in information acquisition, and the formation of matches based on available information. They show that agents reveal their preference truthfully in an equilibrium under some assumptions on agent utility.

The paper proceeds as follows. Section 2 presents a simple example that illustrates why signals can facilitate coordination failure. Section 3 outlines our general model and introduces some notations. Equilibrium analysis is presented in Section 4. Section 5 analyzes the welfare of agents in the model with and without signals. Section 6 compares these welfare implications with the results in the previous literature and discusses two controversial roles of signals in matching markets. A public signaling mechanism is considered in Section 7. Finally, Section 8 discusses some assumptions of our model and concludes.

2 A simple example

Let us consider a market with three firms and three workers. Each firm ranks the workers in the same way (w_1, w_2, w_3) , i.e. each firm strictly prefers worker w_1 to worker w_2 to worker w_3 . Each worker's preference is either typical (f_1, f_2, f_3) with probability $1-\varepsilon$ or atypical with the complementary probability ε , where ε is small. The atypical preferences are independently uniformly distributed among all possible preference order lists. All workers are acceptable to all firms and vice versa.

We first examine behavior in the game in the absence of a signaling mechanism. The only possible match in a sequential equilibrium of the game without signals is the assortative match, in which each firm is matched to the corresponding worker. Let now us analyze the game with the signaling mechanism, described above. We consider the following equilibrium strategies of agents.⁷ Each worker with typical preferences sends her signal to the corresponding firm, i.e. worker w_i sends her signal to firm f_i . Each worker with atypical preferences sends her signal to the best firm worse or equal to the corresponding one (according to typical preferences). Each firm makes its offer to a worker better or equal to the

⁷See Theorem 1 for the proof that these strategies constitute a sequential equilibrium.

corresponding one, only if it receives a signal from her. Each firm ignores all signals from workers worse than the corresponding one. If a firm receives no signals, it makes an offer to the best worker worse than the corresponding one.



Figure I.

Let us consider the realization of preference profiles when only worker w_1 is atypical and firm f_3 is her favorite firm. Worker w_2 and worker w_3 are typical. Figure I illustrates the equilibrium behavior. Worker w_1 sends her signal to firm f_3 . Worker w_2 and worker w_3 send their signals to firm f_2 and firm f_3 correspondingly. Firm f_3 makes an offer to worker w_1 , and firm f_1 anticipates that worker w_1 is atypical and makes an offer to worker w_2 . Firm f_2 also makes its offer to worker w_2 and eventually ends up unmatched because worker w_2 accepts firm f_1 's offer. The coordination failure arises because firm f_2 has no information about worker w_1 's type and cannot anticipate firm f_1 's behavior. Thus, the number of matches for some realization of preferences is smaller than the number of matches when the signals are not allowed. Therefore, the expected number of matches is also smaller.

3 Model

We consider a two-sided matching model with W workers and F firms, $W \ge F$. The set of workers and the set of firms are denoted as W and F correspondingly. Both W and \mathcal{F} include the empty set. Each worker w orders firms according to some strict preference list θ_w . Similarly, each firm f orders workers according to some preference list θ_f . Θ_W and $\Theta_{\mathcal{F}}$ together comprise the set of all possible workers' and firms' preference lists.

Each agent a has cardinal utility compatible with her/its preference list θ_a .⁸ If worker w with preferences θ_w is matched with firm f, she receives cardinal utility $u_w(f, \theta_w)$. Similarly, if firm f with preferences θ_f is matched with worker w, it receives cardinal utility $u_f(w, \theta_f)$. We assume that agent utility depends only on the rank of an agent with which it is matched.

⁸We employ cardinal utilities compatible with ordinal ranking similar to Bogomolnaia and Moulin (2001).

Specifically, the utility of an agent from being matched with an agent on the *kth* position in her/its preference list equals $u_a(k)$. We assume that agents have the same utility function; i.e. for any agent a, $u_a(k) = u(k)$. Our results do not depend on the last assumption; however, this assumption simplifies the exposition.

Additionally, agent's cardinal utility from being unmatched is normalized to zero. We also assume that there is no worker whom firms do not want to hire, and there is no worker who prefers being unemployed to being matched with some firm; i.e. for any k, u(k) > 0.

Each agent knows only her/its preferences and has some ex-ante common beliefs about the other agents' preferences. We consider an environment where each firm f has some fixed publicly known preference list θ_f . Each worker is one of two types: "typical" or "atypical". A "typical" worker w is denoted as w(T). All workers of typical type have the same commonly known preference list θ_0 . An "atypical" worker w is denoted as w(A). The preferences of atypical workers are identically and independently distributed according to some distribution $A(\Theta_W)$. Each worker is ex-ante typical with probability $1 - \varepsilon$ and atypical with probability ε , for some $\varepsilon \in (0, 1)$. Our main analysis considers the case when ε is small.⁹ We also assume that the distribution of atypical preferences, $A(\Theta_W)$, has a full support, i.e. each firm can be the top firm of an atypical worker with positive probability.¹⁰

To model the influence of signals on congested markets, we consider a model with three periods:

- 1. Agents' preferences are realized. Each worker sends a signal to at most one firm; signals are sent simultaneously. Signals are observed only by firms who have received them.
- 2. Each firm makes an offer to at most one worker; offers are made simultaneously.¹¹
- 3. Each worker may accept at most one offer from the set of offers she receives.

We restrict our analysis to pure strategies.¹² A strategy of worker w is a duple $s_w = (s_w^1, s_w^2)$ that represents her decisions at the first and third stages. A strategy of a worker at the first stage is to choose a firm she sends her signal to, $s_w^1 : \Theta_W \to \mathcal{F}$. A strategy of a worker on the last stage is to choose an offer among those available to her, $s_w^2 : \Theta_W \times 2^F \to \mathcal{F}$, where $2^F = \{h : h \subset \mathcal{F}\}$. A strategy of firm f is its decision at the second stage. Firm f chooses the worker to whom it makes an offer based on a set of signals it receives, $s_f : 2^W \to W$, where

⁹The exact bound on ε depends on the parameters of distribution $A(\Theta_{\mathcal{W}})$. However, for each distribution $A(\Theta_{\mathcal{W}})$, one could find an upper bound of ε . We provide a more detailed discussion in Section 8.

¹⁰Formally, for any $f \in \mathcal{F}$ and any $w \in \mathcal{W} \Pr(f = \max_{\theta_w} (f' : f' \in \mathcal{F})) > 0$.

¹¹In practice, some firms should rationally make several offers, anticipating that some workers probably reject their offers. We do not model these strategic decisions.

¹²The analysis of the offer game in which agents can use mixed strategies does not give additional intuition to our main result that signals could impede match formation for some environments. However, this analysis is available upon request.

 $2^W = \{h : h \subset W\}$. The dependence of firm strategy on preferences is omitted, because we assume that each firm has some fixed preferences.

For a given strategy profile of agents $s = (s_w, s_f)$ and realized agents' types $\theta \in (\Theta_W)^W \times (\Theta_F)^F$ one can determine the final matching and agents' utilities. We denote the utility of agent *a* given a strategy profile *s* and a profile of types θ as $\pi_a(s, \theta)$. The interim expected payoff of worker *w* with preferences θ_w from strategy s_w when the other agents follow a strategy profile s_{-w} equals

$$u_w(s_w|s_{-w}, \theta_w) = \sum_{\theta_{-w}} t(\theta_{-w}) \pi_w((s_w, s_{-w}), (\theta_w, \theta_{-w})),$$

where $t(\theta_{-w})$ denotes the joint distribution of all agents except worker w preferences. The interim expected payoff of firm f given a subset of received signals $h \subset \mathcal{W}$, beliefs $\mu_f(\cdot|h)$, and other agents' strategy profile s_{-f} is

$$u_f(s_f|s_{-f},h) = \sum_{\theta} \mu_f(\theta|h) \pi_f(s_f,s_{-f},\theta).$$

We employ the concept of sequential equilibrium for multi-stage games with observed actions and incomplete information in order to solve the game (see Fudenberg and Tirole, 1991). **Definition 1** A strategy profile (s_w, s_f) and posterior beliefs $\mu_f(\cdot|h)$ for each firm f and each subset of workers $h \subset W$ is a sequential equilibrium if

- for any $w \in \mathcal{W}, \ \theta_w \in \Theta_{\mathcal{W}} : s^1_w(\theta_w) \in \arg\max_{\alpha \in \mathcal{F}} u_w(\alpha | s_{-w}, \theta_w),$
- for any $f \in \mathcal{F}$, $h \subset \mathcal{W}$: $s_f(h) \in \arg \max_{\beta \in \mathcal{W}} u_f(\beta | s_{-f}, h)$, and
- for any $w \in \mathcal{W}$, $\theta_w \in \Theta_{\mathcal{W}}$, $h' \subset \mathcal{F} : s_w^2(h', \theta_w) \in \arg \max_{\gamma \in h'} u_w(\gamma, \theta_w)$,

where beliefs are defined using Bayes' rule.¹³

Now we introduce some notations that will be useful in our further discussion. Though worker strategy is a duple $s_w = (s_w^1, s_w^2)$, we will talk mainly about worker strategies at the first stage. The reason is that each worker has a strictly dominant strategy at the last stage-accept the best offer available-since she knows her preferences and the preferences are strict. To simplify notation, we omit the upper index and write $s_w(\theta_w)$ instead of $s_w^1(\theta_w)$.

For convenience, we name firms according to the typical preference list $\theta_0 = (f_1, ..., f_F)$; i.e. f_1 is the best firm, f_2 is the second best, etc. Similarly, we name workers in the following way: worker w_1 is the best worker among all workers \mathcal{W} according to firm f_1 's preferences, $w_1 = \max_{\theta_{f_1}} (w | w \in \mathcal{W})$; worker w_2 is the best worker among $\mathcal{W} \setminus \{w_1\}$ according to firm f_2 's preferences, $w_2 = \max_{\theta_{f_2}} (w | w \in \mathcal{W} \setminus \{w_1\})$; and so on. Generally, worker $w_i = \max_{\theta_{f_i}} (w | w \in \mathcal{W} \setminus \{w_1, ..., w_F\})$ are named according to some prespecified order.¹⁴

We say a subset of workers $h \subset W$ is reached for firm f when workers follow strategy profile s_W if ex-ante probability that only workers from set h send their signals to firm f strictly more than zero.

Definition 2 A subset of workers $h \subset W$ is reached for firm f when workers follow strategy profile s_{W} if

$$\Pr(h_f = h) = \sum_{\theta} t(\theta) \prod_{w \in h} I_{s_w(\theta_w) = f} \prod_{w' \notin h} (1 - I_{s_{w'}(\theta_{w'}) = f}) > 0,$$
where $I_{s_w(\theta_w) = f} = \begin{cases} 1 & \text{if } s_w(\theta_w) = f \\ 0 & \text{otherwise} \end{cases}$ and $t(\theta)$ denotes the joint distribution of all agents, preferences.

We also say that firm f responds to worker w's signal, when workers follow strategy profile $s_{\mathcal{W}}$, if her signal changes the strategy of firm f with positive probability.

¹³Off-equilibrium beliefs are defined by considering the limits of completely mixed strategies.

¹⁴For instance, if all firms have the same preferences θ^* , workers are named according to this preference list $\theta^* = \{w_1, ..., w_W\}$.

Definition 3 Firm f responds to worker w's signal, when workers follow strategy profile s_{W} , if there exists a subset of workers $h, w \notin h$, such that both h and $h \cup w$ are reached for firm f, and $s_f(h) \neq s_f(h \cup w)$.

We proceed with equilibrium analysis in the next section.

4 Equilibrium analysis

As a benchmark, we first consider an environment in which workers cannot send signals. Then, the model outlined above is a static game of incomplete information. Therefore, the notion of sequential equilibrium coincides with the notion of Bayesian equilibrium and agents' beliefs are irrelevant. There is a unique equilibrium match in this case.

If signals are not allowed and ε is small, the only optimal strategy of firm f_1 is to make an offer to its best worker $w_1 = \max_{\theta_{f_1}} (w | w \in \mathcal{W})$. The second top firm anticipates that worker w_1 is likely to accept firm f_1 's offer. Hence, the only optimal strategy of firm f_2 is to make an offer to its best worker among $\mathcal{W} \setminus \{w_1\}$, $w_2 = \max_{\theta_{f_2}} (w | w \in \mathcal{W} \setminus \{w_1\})$ and so on. Workers accept the best available offer. Overall, there is the maximum number of matches, F (since $F \leq W$), in the equilibrium when signals are not allowed.

Proposition 1 (No signaling equilibrium) For sufficiently small ε , there is a unique equilibrium when signals are not allowed: firm f_j , j = 1, ..., F, makes an offer to worker w_j ; worker w_i , i = 1, ..., F, accepts the best available offer.

We further call the match in our benchmark model as "no signaling" match.

Now, we analyze the set of equilibria in the matching market with signals. Though signals are voluntary in our model, they could still play a negative role and draw away firm offers. In order to eliminate such equilibria, we assume that if firm f makes an offer to worker w when it does not receive her signal, firm f makes an offer to worker w when it receives her signal.¹⁵

Assumption PRS (Positive Role of Signals). For any firm $f \in \mathcal{F}$ and any worker $w \in \mathcal{W}$ and any $h \subset \mathcal{W}$, $w \notin h$, if $s_f(h) = w$ then $s_f(h \cup w) = w$.

We further distinguish three types of equilibria in the matching model with signals.

¹⁵See Example A1 in Appendix for an example of an equilibrium in which Assumption PRS is violated.

Definition 4

- An equilibrium is "babbling" if no firm responds to any signal.
- An equilibrium is "informative", if at least one firm responds to some worker's signal.
- An equilibrium is "very informative", if each firm responds to all signals from workers better or equal to its no signaling match.

The set of the first and second type equilibria, i.e. babbling and informative, exhaust the set of all possible equilibria in our model. The set of equilibria of the last type is a subset of the set of informative equilibria.

A babbling equilibrium always exists in our model because signals are costless. If firms do not respond to signals, signals play no role in equilibria. Hence, the only possible match in a babbling equilibrium is no signaling match.

Proposition 2 For sufficiently small ε , the only possible match in a babbling equilibrium is no signaling match.

If some firms respond to signals, then signals transmit information about workers' preferences in an equilibrium, which changes the overall matching outcome. However, there is a great multiplicity of informative equilibria. One may suggest to use refinements proposed by (Cho and Kreps, 1987) and (Banks and Sobel, 1987).¹⁶ However, these criteria are very powerful in the case of one sender and one receiver. The situation with many senders and receivers is more difficult. Though these criteria significantly reduce the number of equilibria, they do not guarantee uniqueness.

However, it is sufficient to restrict ourselves to the case in which each firm responds to all signals from workers better or equal to its no signaling match, i.e. very informative equilibria, in order to guarantee uniqueness. This equilibrium consists of the following strategies. Worker w_i sends her signal to the best firm among the firms that prefer worker w_i to their no signaling match $\Delta(w_i) = (f_j \in \mathcal{F} : w_i \succeq_{f_j} w_j)$. If firm f_j receives at least one signal from the set of workers $\Delta(f_j) = (w \in \mathcal{W} : w \succeq_{f_j} w_j)$, i.e. workers better or equal to worker w_j , it makes its offer to the best such worker; otherwise, it makes an offer to its best worker among $\mathcal{W} \setminus \{w_1, ..., w_j\}$.

¹⁶Cho and Kreps (1987) analyze never a weak responce, intuitive criterion, D1, and D2 refinements. Banks and Sobel (1987) analyze divinity and universal divinity refinements.

Theorem 1 For a sufficiently small ε , under Assumption PRS the set of strategies,

•
$$s_{w_i}(\theta_{w_i}) = \max_{\theta_{w_i}} (f \in \Delta(w_i)),$$

•
$$s_{f_j}(h) = \begin{cases} \max_{\theta_{f_j}}(w : w \in h) & \text{if } h \cap \Delta(f_j) \neq \emptyset \\ \max_{\theta_{f_j}}(w : w \in \mathcal{W} \setminus \{w_1, ..., w_j\}) & \text{if } h \cap \Delta(f_j) = \emptyset \end{cases}$$

and the set of firms' beliefs consistent with agents' strategies constitute a unique very informative equilibrium.¹⁷

The above theorem is remarkable because it shows that the equilibrium of the model is unique, if we restrict our attention to the case in which firms use signals most extensively. However, we should point out that we do not intend to eliminate all other equilibria. First, the theorem illustrates typical agents' behavior in an informative equilibrium. Workers do not just send signals to the best firms. They send their signals to the best firms that respond to these signals, which is in line with AEA advice to participants in the job market for new Ph.D. economists (see AEA, 2005). Similarly, firms do not respond to all signals. Instead they respond to the signals from workers better than those they could secure in the no signaling equilibrium. Second, our results of welfare comparison do hold for most other sequential equilibria.

5 Welfare properties of equilibria

We evaluate the effect of signals on the matching market from an ex-ante perspective. We mainly use the following quantitative characteristics: the expected number of matches, the expected total welfare of firms, and the expected total welfare of workers.

Let us denote the ex-post number of matches for the profile of preferences $\theta \in \Theta_{\mathcal{W}} \times \Theta_{\mathcal{F}}$, when agents follow the profile of strategies s as $m(s, \theta)$. Then, the expected number of matches can be expressed as

$$E[M(s)] = \sum_{\theta} t(\theta) m(s(\theta), \theta),$$

where $t(\theta)$ denotes the joint distribution of all agents' preferences. Similarly, the expected total welfare of workers and firms can be expressed as

$$E[W_{\text{firm}}(s)] = \sum_{f} \sum_{\theta} t(\theta) \pi_{f}(s(\theta), \theta), \text{ and}$$
$$E[W_{\text{worker}}(s)] = \sum_{w} \sum_{\theta} t(\theta) \pi_{w}(s(\theta), \theta)$$

 $^{^{17}}$ We should point out that there is a multiplicity of beliefs that could support this equilibrium on offequilibrium path.

correspondingly.

Proposition 1 shows that the expected number of matches in any no signaling equilibrium is the maximum one. Hence, it is impossible that the expected number of matches in any informative equilibrium exceeds the expected number of matches in any "no signaling" equilibrium. Example 1 and Example 2 demonstrate the case of strict inequality and equality for this welfare criterion.

Example 1 is presented in the introduction and considers the very informative equilibrium with three firms and three workers. To avoid a repetition, we do not discuss it here.

Example 1 (Fewer expected number of matches) There are three firms and three workers .Firms have the same ranking over workers (w_1, w_2, w_3) . The typical worker preference list is $\theta_0 = (f_1, f_2, f_3)$. Atypical worker preferences are uniformly distributed. Firm f_j , j = 1, 2, 3, and worker w_i , i = 1, 2, 3, equilibrium strategies are

• $s_{w_i}(\theta_{w_i}) = \max_{\theta_{w_i}} (f \in \Delta(w_i)),$

•
$$s_{f_j}(h) = \begin{cases} \max_{\theta_{f_j}}(w : w \in h) & \text{if } h \cap \Delta(f_j) \neq \emptyset \\ \max_{\theta_{f_j}}(w : w \in \mathcal{W} \setminus \{w_1, ..., w_j\}) & \text{if } h \cap \Delta(f_j) = \emptyset \end{cases}$$

and the set of firms' beliefs consistent with agents' strategies.

Example 2 shows that some informative equilibria could have the maximum expected number of matches. Intuitively, it is possible that if some firm f_j secures a better match with some atypical worker w_i , firm f_i always makes its offer to firm f_j 's no signaling match, worker w_j , in an equilibrium. Therefore, firms exchange their matches and it does not decrease the number of matches.

Example 2 (Equal expected number of matches) Let us consider three firms and three workers. All firms have the same preferences $\theta_{f_j} = \{w_1, w_2, w_3\}$. Let us consider the following equilibrium strategies:

•
$$s_{w_1}(\theta_{w_1}) = \max_{\theta_{w_1}} (f : f \in \{f_1, f_2\}) \text{ and } s_{w_i}(\theta_{w_i}) = f_i, i = 2, 3;$$

• $s_{f_j}(h) = \begin{cases} \max_{\theta_{f_j}} (w : w \in h) & \text{if } h \cap \Delta(f_j) \neq \emptyset \\ \max_{\theta_{f_j}} (w : w \in \mathcal{W} \setminus \{w_1, \dots, w_j\}) & \text{otherwise} \end{cases}$, for $j = 1, 2;$
• $s_{f_3}(h) = \begin{cases} \max_{\theta_{f_3}} (w : w \in h) & \text{if } h \cap \Delta(f_3) \neq \emptyset \\ w_3 & \text{otherwise} \end{cases}$.

The set of equilibrium beliefs is such that if firm f_1 or f_2 receives a signal from worker w_1 , it believes that it is worker w_1 's top firm. If firm f_3 receives a signal from worker w_1 , its belief coincides with her prior, i.e. worker w_1 is typical with probability $1 - \varepsilon$ and atypical with probability ε . Similarly, if any firm f_j receives a signal from worker w_2 or w_3 , its belief coincides with its prior. To put it briefly, only firm f_1 and firm f_2 respond to worker w_1 's signal. All other signals are ignored. One may check that the described strategies constitute an informative equilibrium.

The results about the expected total welfare of firms and the expected total welfare of workers are not so straightforward and depend on the relative magnitudes of u(k). The intuition is that signals in an informative equilibrium play two roles. On the one hand, signals help to secure "better" matches between some atypical workers and firms. On the other hand, the introduction of signals leaves some workers and firms unmatched. Example 3 illustrates that the introduction of signals is beneficial for a matching market according to egalitarian welfare criterion if and only if the decrease in the number of matches is offset by better matches of atypical workers. A similar example can show that the total welfare of firms changes ambiguously.

Example 3 (Welfare of firms and workers) Let us again consider the game of Example 1. Workers' cardinal utilities from being matched to first, second, and third choice are $\delta + \lambda$, δ , and $\delta - \lambda$ correspondingly. The expected total welfare of workers in no signaling match

$$E[W_{worker}^{nosignals}] = \sum_{i=1}^{3} \left[(1-\varepsilon) u(i) + \varepsilon_{\frac{1}{3}} \sum_{l=1}^{3} u(l) \right] = 3\delta.$$

One may check that the expected total welfare of workers in very informative equilibrium is^{18}

$$E[W_{worker}^{signals}] = 3\delta + \left(-\frac{1}{3}\delta + \frac{19}{6}\lambda\right)\varepsilon$$

Hence, the expected total welfare of workers increases, if and only if the difference in utilities between adjacent firms is large enough, $\lambda > \frac{2}{16}\delta$.

The theorem below summarizes the results derived above.

Theorem 2 For a sufficiently small ε :

- the expected number of matches in any informative equilibrium is weakly fewer than in any no signaling equilibrium;
- the effect of signals on the expected total welfare of firms and the expected total welfare of workers is ambiguous.

¹⁸Terms of the order of ε^2 and ε^3 are ignored.

We have compared above the properties of any informative and no signaling equilibrium. However, more strict result holds for very informative equilibrium. Under the assumption that there are at least three workers and there exists a worker w, such that there are at least three firms that weakly prefer worker w to their no signaling matches, $|\{f_j \in \mathcal{F} : w \succeq_{f_j} w_j\}| \geq 3$, the expected total number of matches is strictly fewer in very informative equilibrium than in the corresponding no signaling match. The intuition for this result is similar to the one for Example 1. If firms respond to signals, some of the realized matches differ from no signaling match. Moreover, if at least three firms respond to some worker's signal the exchange of matches—the situation presented in Example 2—is impossible for each realization of preferences. Therefore, the expected number of matches is smaller than the maximum one in this case.

Theorem 3 For sufficiently small ε , if there are at least three workers and for some worker $w, |\Delta(w)| \ge 3$, the expected number of matches is strictly smaller in the very informative equilibrium than in the corresponding no signaling equilibrium.

Theorem 2 proves that the expected total welfare of workers changes ambiguously with the introduction of signals. However, the following proposition shows that signals are harmful to workers only because they deprive them of matches. Workers receive weakly better offers conditional on the fact that they receive any offer.

Proposition 3 If a worker receives an offer in any informative equilibrium, this offer is from a firm weakly better than her no signaling match.

It is easy to see that the above statement is not true for firms, because some firms may have to make offers to workers worse than their no signaling match if she is atypical.

6 Role of signals in matching markets

Coles et al. (2009) show that the introduction of signals increases the expected number of matches and the welfare of workers. However, they consider an environment where agents' preferences are block-uniform. Specifically, there exists a partition $\mathcal{F}_1, \ldots, \mathcal{F}_B$ of the firms into *blocks* and

- 1. For any b < b', where $b, b' \in \{1, \ldots, B\}$, each worker prefers every firm in block \mathcal{F}_b to any firm in block $\mathcal{F}_{b'}$;
- 2. Each worker's preferences within block \mathcal{F}_b are uniform and uncorrelated, for all b;

3. Firm preferences over workers are uniform and uncorrelated.

This paper shows that Coles et al. (2009) results rely on the assumption that preferences are block-uniform. If the preferences of workers are almost aligned and the preferences of firms are fixed and commonly known, the introduction of signals decreases the expected number of matches. The effect of signals on the expected total welfare of agents is ambiguous. Overall, Table I presents the effects from the introduction of the signals for the two different environments: almost complete (this paper) and block-uniform distribution of preferences.

Preferences	No signals	Matches	$E[W_{\rm worker}]$	$E[W_{\rm firm}]$
Almost complete	0	_	±	±
Block-uniform	0	+	+	±

Table I. Almost complete VS Block-uniform distribution preferences.

A natural question is why signals influence matching markets in different ways. We argue that the signals play two different roles: transmit information and facilitate information asymmetry. On the one hand, the introduction of signals helps atypical workers to transmit information about their preferences and locate a better match. On the other hand, signals transmit information only to some firms, thus facilitating information asymmetry. This information asymmetry leads to coordination failures that decrease the number of matches.

When there is ex-ante small amount of information about agents' preferences, information transmission plays a more important role in match formation. This happens when agents' preferences are ex-ante block-uniform, as in Coles et al. (2009). However, when there is almost complete information about agents' preferences—as in the model of this paper—the introduction of signals leads to coordination failures. Table II presents the comparison.

Preferences	Transmit information	Facilitate information asymmetry
Almost complete	Small	Large
Block-uniform	Large	Small

Table II. The roles of signals

Overall, the signals play controversial roles in the match formation process. This could make them a less powerful tool than it was previously anticipated.

7 Public signals

One could conjecture that should signals be public, they would always benefit match formation. Public signals introduce no asymmetry of information among firms. Firms have the same beliefs about the distribution of workers' preferences and the same beliefs about the strategies other firms use. Therefore, firms should be able to make offers that are more likely to be accepted. Unfortunately, this intuition is incorrect. This section illustrates that the expected number of matches in an equilibrium of the offer game with public signals could be smaller than the expected number of matches in the offer game without signals.

We consider a market with three firms and three workers. The distribution of agents' preferences is the same as in Section 3. Each worker can send at most one signal and accept at most one offer. Each firm has only one vacant position and can make at most one offer. The timing of the game is as follows:

- 1. Agents' preferences are realized. Each worker sends a signal to at most one firm; signals are sent simultaneously. All agents observe what signals each firm receives.
- 2. Each firm makes an offer to at most one worker; offers are made simultaneously.
- 3. Each worker chooses an offer to accept from the set of offers she receives.

The only difference from the game we considered previously is that all agents observe the signals each firm receives. The strategies of workers are the same as in Section 3. However, a strategy of firm f now depends on the set of signals each firm receives, $s_f : \mathcal{F}^W \to \mathcal{W}$.¹⁹

As previously, the only equilibrium outcome of the offer game with signals is a full match. However, the expected number of matches could be smaller than three if we allow workers to send public signals. Intuitively, public signals do not transmit enough information about workers' preferences. This could introduce a considerable amount of uncertainty about workers' preferences. Therefore, some firms can optimally engage in a competitive behavior for some workers; i.e. firms make their offers to the same worker in an equilibrium. This produces mismatches.

Example 4 There are three firms and three workers. Firms have the same ranking over workers, which we denote as (w_1, w_2, w_3) . The typical worker preference list is (f_1, f_2, f_3) . The atypical worker preferences are uniformly distributed. We assume that all firms have the same cardinal utility and their utility from being matched to second top worker, i.e. u(2), is at least twice as great as the cardinal utility from being matched to the third top worker, i.e. u(3).

 $^{^{19}}$ Note that we again omit the dependence of strategies on firms' preferences, as we assume that each firm has fixed commonly known preferences.

We consider the following strategies of agents in the offer game with public signals. Worker w_i sends her signal to the best firm among the firms that weakly prefer worker w_i to their no signaling match $\Delta(w_i) = (f_j \in \mathcal{F} : w_i \succeq_{f_j} w_j)$:

$$s_{w_i}(\theta_{w_i}) = \max_{\theta_{w_i}} (f \in \Delta(w_i)).$$

Firms use the following strategies.

- 1. Firm f_1 makes an offer to worker w_1 , if it receives a signal from her; otherwise, it makes an offer to worker w_2 .
- 2. Firm f_2 makes an offer to worker w_1 , if it receives a signal from her. Firm f_2 makes an offer to worker w_3 , if either worker w_1 sends a signal to firm f_1 and worker w_2 sends a signal to firm f_3 or worker w_1 sends a signal to firm f_3 and worker w_2 sends a signal to firm f_2 . In all other cases, firm f_2 makes an offer to worker w_2 .
- 3. Firm f_3 makes an offer to the best worker from whom it receives a signal. If it receives no signals, it makes an offer to worker w_3 .

Each firm's beliefs on the equilibrium path are consistent with agents' strategies and each firm off-equilibrium beliefs coincide with priors.

Let us consider the strategies outlined in Example 4. Mismatches happen when both worker w_1 and worker w_2 are atypical. If at least two atypical workers send their signals to the same firm, only one worker receives an offer from it. Since, signals are public, all other firms infer that the other worker is atypical. This creates a considerable amount of uncertainty about the worker preferences. Since, this worker could be a good one, firms have incentives to compete for her.

Another reason for excessive competition among firms is that signals may not transmit information about workers' top firms. Some workers send their signals to firms that differ from their top ones in an equilibrium, because they want to attract feasible offers. Therefore, several firms could have incentives to make an offer to a given worker. This creates competition among firms, which again lead to mismatches.

Proposition 4 formally proves that the set of strategies in Example 4 constitutes a sequential equilibrium.

Proposition 4 The set of strategies in Example 4 constitutes a sequential equilibrium.



Figure 1: Figure II. Public signals.

The implications of the above example can be summarized by way of two observations. First, public signals do not transmit enough information about workers' preferences. This could introduce uncertainty about workers preferences and induce excessive firm competition for the same workers. This results in mismatches.

In addition, mismatches in the offer game with public signals occur only if there at least two atypical workers, which happens only with probability of the order ε^2 . In contrast, mismatches in the offer game with private signals occur with the probability of the order ε . Therefore, mismatches happen less often when signals are public.

8 Conclusion

There is a general belief that preference signaling should facilitate match formation (see Crawford and Sobel, 1982; Roth, 2008; AEA, 2005). This belief is also supported by Coles et al. (2009) who show that the introduction of signals increases the expected number of matches and welfare of workers. We show in this paper that this belief can be erroneous for some matching markets. We exemplify an environment in which the introduction of signals harms matching markets: it decreases the expected number of matches and ambiguously affects the welfare of firms and the welfare of workers. Based on this example, we argue that signals play controversial roles in match formation. Though they help to transmit information about participants' preferences, they also facilitate information asymmetry among them. While the former effect reduces coordination failures and facilitates better match formation, the latter effect acts in the opposite direction. Finally, we show that making signals observable to all agents does not change the welfare applications. We leave here as an open empirical question which effect dominates in real matching markets.

In conclusion, we want to discuss some assumptions of our model. We analyze the introduction of signals to congested decentralized matching markets, as we believe the job market for new Ph.D. economists to be. The fact that we do not analyze any centralized clearinghouse mechanism and stable matches captures the idea of decentralized markets. The fact that we analyze a one-period model captures the idea of congestion. Moreover, several (but finite) periods of interactions between firms and workers would give an opportunity for firms to secure better matches; but signals would still introduce information asymmetry. If each worker sent several signals, these would transmit information to a greater number of firms, but each signal would be less informative. Several vacant positions would make only the preferences of firms more complicated and would not influence the results. Overall, the roles of signals in match formation are robust to these modifications.

The results of this paper are in terms of a sufficiently small ε . However, what we really need is the uniqueness of equilibrium in the benchmark model without signals. The multiplicity of equilibria does not allow a clean comparison between models with and without signals. One could check that there is a unique equilibrium in no signaling model with uniform distribution of atypical preferences if $\varepsilon < \min(\min_i(\frac{u(i)-u(i+1)}{u(i)}), \frac{u(F)}{u(F)+0.5u(1)})$.

We should also highlight that we rely on cardinal utility in our analysis. Our results could not be extended to an ordinal framework, because an Ordinal Bayesian Nash Equilibrium (see d'Aspremont and Peleg, 1988) may not exist in our environment.

A Appendix

Proof of Proposition 1.

Let us show this statement for each firm sequentially. Each worker w_i has preferences $\theta_0 = (f_1, ..., f_F)$ with probability $1 - \varepsilon$ and with complementary probability some preferences distributed according to $A(\Theta_W)$. Let us consider firm f_1 which has some preferences θ_{f_1} . If it makes an offer to worker $w_1 = max_{\theta_{f_1}}(w|w \in W)$, its offer will be the best worker w_1 's offer with probability at least $1 - \varepsilon$. Hence, its expected utility from making an offer to worker w_1 equals at least $(1 - \varepsilon)u(1)$ which is greater than u(2) for sufficiently small ε . Hence, independently on other firms' strategies, firm f_1 's optimal strategy is to make an offer to its best worker.

Let us assume that each firm f_k , k < j, makes its offer to worker w_k . Now we consider the decision of firm f_j . The expected payoff from making an offer to some worker among $\{w_1, ..., w_{j-1}\}$ is less than $\varepsilon u(1)$. In the same time the expected payoff from making offer to some worker among $\mathcal{W} \setminus \{w_1, ..., w_{j-1}\}$ is at least $(1 - \varepsilon)u(j)$. Hence, given the strategies of other firms and sufficiently small ε , the optimal strategy of firm f_j is to make an offer to its best worker among $\mathcal{W} \setminus \{w_1, ..., w_{j-1}\}$. \Box

Proof of Proposition 2.

The only undominated strategy of a worker at the last stage is to choose the best offer among available ones. Then, under the condition that firm f does not respond to any signal, for any $h \subset W$ reached in an equilibrium $s_f(h) = const$. Let us assume that there exists a realization of agents' preferences such that firm f_1 is matched to some worker w_i , i > 1, in the equilibrium. Hence, for any $h \subset W$, reached in the equilibrium, $s_{f_1}(h) = w_i$. Hence, the expected firm 1's payoff equals at most u(2). However, the strategy $s_{f_1}(h) = w_1$ for any $h \subset W$ is compatible with assumption that firm f_1 does not respond to any signals and gives payoff $(1 - \varepsilon) u(1)$ independently of strategies of other firms. Hence, $s_{f_1}(h) = w_i$ cannot be an equilibrium strategy. Similar argument could be applied to any other firm f_j , $j = 2, ..., F.\Box$

Proof of Theorem 1.

We prove the theorem by way of several lemmata. In the proof of the lemmata we presume that ε is sufficiently small. First, we establish that a firm believes about a particular worker is typical with probability more than $1 - \varepsilon$ either when it receives her signal or when it does not receive her signal. Second, we show that firms do not make their offers to a worker better than no signaling match if they do not receive her signal. The third lemma proves that if a firm does receive a signal from a worker better than its no signaling match, it makes its offer to the best such worker. Finally, using the statements of lemmata we show that the set of strategies stated in the theorem constitutes a unique very informative equilibrium.

First two lemmata do not require the assumption that each firm f_j , j = 1, ..., F, responds to all signals from workers better or equal to worker w_j according to its preferences.

Lemma A1 For any worker $w \in \mathcal{W}$, any firm $f \in \mathcal{F}$, and any $h \subset \mathcal{W}$ either $\mu_f(\theta_w = \theta_0 | h \cup w) \ge 1 - \varepsilon$ or $\mu_f(\theta_w = \theta_0 | h \setminus w) \ge 1 - \varepsilon$. Similarly, either $\mu_f(\theta_w \neq \theta_0 | h \cup w) \le \varepsilon$ or $\mu_f(\theta_w \neq \theta_0 | h \setminus w) \le \varepsilon$.

Proof.

Let us denote as α_T and α_A the probabilities that typical and atypical type of worker w correspondingly send a signal to firm f. Then, if worker w sends her signal to firm f, $(1 - \varepsilon)\alpha_T + \varepsilon \alpha_A > 0$, we derive its beliefs using Bayes' rule

$$\begin{cases} \mu_f(\theta_w = \theta_0 | h \cup w) = \frac{(1-\varepsilon)\alpha_T}{(1-\varepsilon)\alpha_T + \varepsilon\alpha_A} \\ \mu_f(\theta_w = \theta_0 | h \setminus w) = \frac{(1-\varepsilon)(1-\alpha_T)}{(1-\varepsilon)(1-\alpha_T) + \varepsilon(1-\alpha_A)} \end{cases}$$

One can verify that

$$\begin{cases} \mu_f(\theta_w = \theta_0 | h \cup w) \ge 1 - \varepsilon \iff \alpha_T \ge \alpha_A \\ \mu_f(\theta_w = \theta_0 | h \backslash w) \ge 1 - \varepsilon \iff \alpha_T \le \alpha_A \end{cases}$$

Hence, either $\mu_f(\theta_w = \theta_0 | h \cup w) \ge 1 - \varepsilon$ or $\mu_f(\theta_w = \theta_0 | h \setminus w) \ge 1 - \varepsilon$. If worker w never sends her signal to firm f, $(1 - \varepsilon)\alpha_T + \varepsilon \alpha_A = 0$, firm f's beliefs are $\mu_f(\theta_w = \theta_0 | h \setminus w) = 1 - \varepsilon$ and $\mu_f(\theta_w = \theta_0 | h \cup w)$ is arbitrary. The second statement directly follows from the first one. \Box

Lemma A2 (Offer to better workers) If firm f_j does not receive a signal from worker w strictly better than worker w_j , $w \succ_{f_j} w_j$ it does not make an offer to her in an equilibrium.

Proof.

We prove this statement for firms sequentially. Let us first show its validity for j = 2. The only worker that could be better than worker w_2 for firm f_2 is worker w_1 by construction. If $w_2 \succ_{f_2} w_1$ we are done. Assume that $w_1 \succ_{f_2} w_2$.

There are two possibilities: either worker $w_1(T)$ sends her signal to firm f_1 , i.e. $s_{w_1}(\theta_0) = f_1$, or she does not send her signal to firm f_1 , i.e. $s_{w_1}(\theta_0) \neq f_1$, in an equilibrium.

Assume worker w_1 employs strategy $s_{w_1}(\theta_0) = f_1$. If firm f_2 does not receive worker w_1 signal, firm f_2 believes she is atypical with probability less than ε , $\mu_{f_2}(\theta_{w_1} \neq \theta_0 | h \setminus w_1) \leq \varepsilon$ (Lemma A1). According to assumption $F \leq W$, firm f_2 can secure a match with some worker $w_i, i \geq 2$, with probability at least $1 - \varepsilon$. Hence, firm f_2 does not make an offer to worker w_1 in an equilibrium.

Worker w_1 employs strategy $s_{w_1}(\theta_0) \neq f_1$ in an equilibrium only if firm f_1 makes its offer to worker w_1 with probability equals to one, and firm f_2 has a chance to be matched with worker w_1 only if she is atypical. Assume firm f_2 makes an offer to worker w_1 when it does not receive her signal. If $w_1(T)$ sends her signal to firm f_2 in an equilibrium, according to Assumption *PRS* firm f_2 should also make an offer if it receives a signal from w_1 . However, if it receives a signal from w_1 , the probability that worker w_1 is atypical less than ε (Lemma A1), which contradicts equilibrium behavior.

Now, we assume that worker $w_1(T)$ does not send her signal to firm f_2 in an equilibrium. If firm f_2 does not receive worker w_1 's signal, firm f_2 believes that she is atypical with probability less or equal ε , $\mu_{f_2}(\theta_{w_1} \neq \theta_0 | h \setminus w_1) \leq \varepsilon$ (Lemma A1). Therefore, it is again suboptimal for firm f_2 to make an offer to worker w_1 if it does not receive a signal from her.

We have shown above that it is suboptimal for firm f_2 to make an offer to worker w_1 if it does not receive a signal from her. Let us assume that it is suboptimal for any firm f_j , j < k to make its offer to a worker w_t , t < j, if firm f_j does not receive a signal from it and show that the claim for firm f_k .

We consider some worker w_i , i < k. Firm f_i makes its offer to workers $\{w_1, ..., w_{i-1}\}$ with probability less than $\varepsilon (i-1)$. In addition, worker w_i is atypical with probability ε . Hence, firm f_k can secure a match with worker w_i with probability equals at most $i\varepsilon$ if it does not receive a signal from her. For small enough ε firm f_k 's offer to worker w_i is suboptimal. \Box

Now, we assume that each firm f_j , j = 1, ..., F, responds to all signals from workers better or equal to worker w_j according to its preferences. The following lemma shows that firm f_j makes its offer to some worker w better or equal to worker w_j if worker w's signal is the best signal firm f_j receives. Lemma A3 (Response to signals) Assume that F > W. Then, for any $h \subset W$ $s_{f_j}(h) = \max_{\theta_{f_j}} (w : w \in h)$ if $h \cap \Delta(f_j) \neq \emptyset$ in very informative equilibrium²⁰.

Proof.

We prove this statement for firms sequentially. Let us consider firm f_1 and worker w_1 . Assume that worker w_1 employs strategy $s_{w_1}(\theta_0) \neq f_1$. Then, firm f_1 believes that for any $h \subset \mathcal{W} \ \mu_{f_1}(\theta_{w_1} = \theta_0 | h \setminus w_1) \geq 1 - \varepsilon$. Therefore, for sufficiently small ε , firm f_1 always makes its offer to worker w_1 , which contradicts to our assumption that it responds to worker w_1 's signal. Therefore, under the assumption that firm f_1 responds to a signal from worker w_1 , the only possible worker w_1 's equilibrium strategy is $s_{w_1}(\theta_0) = f_1$. In this case, for any $h \subset \mathcal{W}$ firm f_1 's belief is $\mu_{f_1}(\theta_{w_1} = \theta_0 | h \cup w_1) \geq 1 - \varepsilon$. Hence, firm f_1 's highest expected payoff when it receives worker w_1 's signal is from making an offer to worker w_1 . Hence, for any $h \subset \mathcal{W}$, firm f_1 's strategy $s_{f_1}(h \cup w_1) = w_1$ is optimal.

Assume now that for any $t \leq j < k$, and for any $h \subset \mathcal{W}$, firm f_j employs strategy for $s_{f_j}(h) = \max_{\theta_{f_j}}(w : w \in h)$ if $h \cap \Delta(f_j) \neq \emptyset$. We prove below that firm f_k 's optimal strategy for any $h \subset \mathcal{W}$ and $s_{f_k}(h) = \max(w : w \in h)$ if $h \cap \Delta(f_k) \neq \emptyset$.

There are two possibilities: either $s_{w_k}(\theta_0) \neq f_k$ or $s_{w_k}(\theta_0) = f_k$. For the former case, for any $h \in \mathcal{W} \ \mu_{f_k}(\theta_{w_k} = \theta_0 | h \setminus w_k) \geq 1 - \varepsilon$. Hence, it is optimal for firm f_k to make an offer to worker w_k when it receives no signals from workers better or equal to worker w_k , i.e. for any $h' \in \mathcal{W}$ such that $h' \cap \Delta(f_k) = \emptyset$, $s_{f_k}(h') = w_k$. Hence, it is also optimal for firm f_k to make an offer to worker w_k when worker w_k 's signal is the best signal it receives, i.e. for any $h'' \in \mathcal{W}$ such that such that $h'' \cap \Delta(f_k) = w_k$, $s_{f_k}(h'') = w_k$. Therefore, firm f_k does not respond to worker w_k 's signal. Contradiction.

For the latter case, $s_{w_k}(\theta_0) = f_k$, if firm f_k does not receive a signal from worker w_k , it anticipates that she is atypical. Therefore, firm f_k does not make its offer to her. If firm f_j receives signals from any worker $w_i \succeq w_k$ no other firm f_p , $p \neq j$ and p > i, makes its offer to worker w_i according to Lemma A2. The only offers that compete with firm f_j 's offer could be the ones from the set $\{f_p, p < i\}$. However, any firm f_p , p < i, could make an offer to worker w_i only if worker w_p is atypical, which happens with probability ε . Hence, the interim expected payoff for firm f_j from making its offer to worker w_i equals at least $(1 - (i - 1)\varepsilon)u'$, where $u' = u_{f_j}(w_i, \theta_{f_j})$. Firm f_j expected payoff from making an offer to any other worker from set $\Delta(f_j)$ is smaller than $(1 - (i - 1)\varepsilon)u'$ as this worker either has not sent a signal to firm f_j or has a smaller rank in firm f_j' preferences. The expected payoff from making an offer to some worker $W \setminus \Delta(f_j)$ is smaller either. Therefore, firm f_j optimal strategy is, $s_{f_j}(h) = max_{\theta_{f_j}}(w: w \in h)$ if $h \cap \Delta(f_j) \neq \emptyset$. \Box

²⁰If F = W the claim is still valid with the same assumption for all firms except firm f_F . Firm f_F should respond to a signal from any worker strictly better than the corresponding one.

Now we are ready to prove the theorem. Let us show that the set of strategies, stated in the theorem, constitutes an equilibrium. We first prove that if all agents, except firm f_l , follow the strategies, stated in the theorem, firm f_l 's strategy is optimal given its belief is consistent with the other agents' strategies. If firm f_l receives a signal from worker $w_t, t < l$, firm f_l believes that itself is the best firm among $\Delta(w_t) = \{f_j \in \mathcal{F} : w_t \succeq_{f_j} w_j\}$. Let us assume that worker w_t is the best worker who sends a signal to firm f_l . Worker w_t does not accept firm f_l 's offer only if she receives an offer from some firm $f_k \in \mathcal{W} \setminus \Delta(w_t)$. However, it happens only if worker w_k is atypical, i.e. with probability less than ε . Hence, firm f_l interim expected payoff from making an offer to worker w_t equals at least $(1 - (n - 1)\varepsilon) u'$, where $u' = u_{f_l}(w_t, \theta_{f_l})$. Firm f_l 's offer to a worker better than worker w_t is not optimal according to Lemma A1. Firm f_l 's expected payoff from making an offer to some worker $w, w_t \succ_{f_l} w$, is also smaller than making an offer to worker w_t for sufficiently small ε . Overall, firm f_l 's strategy is optimal.

Let us show that, if all agents, except worker w_t , follow the strategies, stated in the theorem, worker w_t 's strategy is optimal. Firm f_t does not make an offer to worker w_t when it receives a signal from a better worker. Therefore, if worker w_t is typical, her payoff from sending a signal to firm f_t equals at least $[1 - (l - 1)\varepsilon] u(t)$. If worker w_t does not send her signal to firm f_t it loses her offer and she could get payoff at most u(t - 1), which is less than $[1 - (l - 1)\varepsilon] u(t)$ for sufficiently small ε . There is also no reason for worker w_t to send her signal to a firm better than firm f_t , because this firm does not respond to her signal according to its equilibrium strategies. Hence, worker $w_t(T)$'s strategy is optimal. Using similar logic one can show that worker $w_t(A)$'s strategy is also optimal.

Now we show that the above strategies constitute the unique very informative equilibrium. Lemmata A2 and A3 imply that each firm f_l , l = 1, ..., F, has to follow the following strategies in an equilibrium:

for any
$$h \subset \mathcal{W}, \begin{cases} s_{f_l}(h) \neq w_l \text{ if } h \cap \Delta(f_l) = \varnothing \\ s_{f_l}(h) = w_l \text{ if } h \cap \Delta(f_l) = w_l \end{cases}$$

Straightforwardly, the only worker $w_l(T)$'s optimal strategy is to send her signals to firm f_l , $s_{w_l}(\theta_0) = f_l$, otherwise, firm f_l 's does not make an offer to student w_l .

Let us consider firm $f^* = \max_{\theta_{w_l}} (f' \in \Delta(w_l))$. Firm f^* responds to signals from workers better or equal than no signaling match and its equilibrium beliefs are $\mu_{f^*}(\theta_{w_l} = \theta_0 | h \setminus w_l) \ge$ $1 - \varepsilon$ and $\mu_{f^*}(\theta_{w_l} \neq \theta_0 | h \cup w_l) = 1$. Therefore, if firm f^* does not receive a signal better than worker w_l 's one, it's optimal strategy is to make an offer to worker w_l . Taking into account that firm f^* can receive a signal from a better worker with probability less than $(l - 1)\varepsilon$, worker $w_l(A)$'s optimal strategy is to send her signal to firm f^* (for sufficiently small ε). Hence, the strategies, stated in the theorem, constitute the unique equilibrium. \Box

Proof of Theorem 3.

Assumption that $A(\Theta_W)$ has a full support and that the strategies of the very informative equilibrium guarantee that some worker w_i sends her signals to each firm in the set $\Delta(w_i) =$ $|\{f_j \in \mathcal{F} : w_i \succeq_{f_j} w_j\}|$ with positive probability. Then, using logic of Example 1 and Example 2 it is straightforward to show that when there are at least three firms in the set $\Delta(w_i)$ and there are at least three workers the mismatch happens with positive probability. Therefore, the expected number of matches strictly smaller than in the corresponding no signaling equilibrium. \Box

Proof of Proposition 3. The statement directly follows from the strategies of the very informative equilibrium. \Box

Example A1 (An equilibrium when assumption PRS does not hold) Let us consider two firms and two workers. We assume that all firms have the same preferences over workers $\theta_{f_1} = \theta_{f_2} = \{w_1, w_2\}$. Also we assume that each typical worker has preferences $\theta_0 = (f_1, f_2)$ and each atypical worker has preferences $\theta_A = (f_2, f_1)$ with probability equal to one. Firms prefer worker w_1 to worker w_2 . Agents employ the following strategies:

 $s_{w_1}(\theta_0) = f_2, \ s_{w_1}(\theta_A) = f_1$ $s_{w_2}(\theta_0) = f_1, \ s_{w_2}(\theta_A) = f_2$ $for \ any \ h \subset \mathcal{W} \ s_{f_1}(h) = \begin{cases} w_1 \ if \ w_1 \notin h \\ w_2 \ if \ w_1 \in h \end{cases}, \ s_{f_2}(h) = \begin{cases} w_1 \ if \ w_1 \notin h \\ w_2 \ if \ w_1 \in h \end{cases}$

Agents' believes are:

- for any $h \subset \mathcal{W} \ \mu_{f_j}(\theta_{w_i} : f_j = \max_{\theta_{w_i}} (f \in \mathcal{F}) | h \setminus w_i) = 1$ and $\mu_{f_j}(\theta_{w_i} : f_j = \min_{\theta_{w_i}} (f \in \mathcal{F}) | h \cup w_i) = 1$

It is easy to show that the above strategies and the set of beliefs constitute a sequential equilibrium. One may extend this example for the environment with more firms and workers.

Proof of Proposition 4.

Let us first prove that firms' strategies are optimal. Note that if firm f receives a signal from worker w_1 it believes that it is her top firm. Therefore, it is optimal for her to make her an offer. Now, if firm f_1 that does not receive a signal from worker w_1 , firm f_1 believes that worker w_1 is atypical and will not accept its offer. Then, firm f_1 strategy of making an offer to worker w_2 is optimal for any signaling pattern, because it believes that her offer will be accepted at least with probability $\frac{1}{2}$. The worst case is when worker w_2 is atypical and sends her signal to firm f_3 .

Now we consider firm f_2 optimal strategy. Let us consider the case when worker w_1 sends her signal to firm f_1 and worker w_2 sends signal to firm f_3 . Firm f_2 believes worker w_2 prefers firm f_3 to itself. Since, firm f_3 makes an offer to worker w_2 , and firm f_1 makes an offer to worker w_1 , the only worker that could accepts firm f_2 offer is worker w_3 . If worker w_1 sends he signal to firm f_3 and worker w_2 sends her signal to firm f_2 . In this case firm f_1 makes an offer to worker w_2 , who is typical with probability $(1 - \frac{1}{3}\varepsilon)$. Since, worker w_1 most preferred firm is firm f_3 , the optimal strategy of firm f_2 to make an offer to worker w_3 .

Let us consider the case firm f_3 receives signals from all workers. In this case firm f_3 makes an offer to worker w_1 . It is optimal for firm f_1 and firm f_2 to make an offer to worker w_2 because her preferences over these firms could be equally likely. Hence, the payoff from making an offer to worker w_2 equal $\frac{1}{2}u(2) > u(3)$. Similar, one could show that in other cases it is optimal for firm f_2 to make an offer to worker w_2 . In a similar way one could show that it is always optimal for firm f_3 to make an offer to the best worker it receives a signal from.

Let us now show each worker uses optimal strategy. Worker w_1 strategy is optimal, because any firm makes her an offer upon receiving her signal. There is no incentive for worker w_2 to make an offer to firm f_1 since, all firms upon observing such behavior has believes about workers preferences that coincides with the priors. Therefore, worker w_2 optimal strategy is to send her signal to the best firms among f_1 and f_2 . Since firms do not put attention to worker w_3 signals, there is no reason for her to deviate from the equilibrium strategy. \Box

References

- Abdulkadiroglu, A., Che, Y.-K., and Yasuda, Y. (2008). Expanding 'choice' in school choice. Working papers, Duke University, Columbia University, and National Graduate Institute for Policy Studies.
- AEA (2005). Signaling for interviews in the economics job market. Technical report, American Economic Association. http://www.aeaweb.org/joe/signal/signaling.pdf.
- Avery, C. and Levin, J. (2009). Early admissions at selective colleges. *American Economic Review*.
- Banks, J. and Sobel, J. (1987). Equilibrium selection in signaling games. *Econometrica*, pages 647–661.
- Bogomolnaia, A. and Moulin, H. (2001). A new solution to the random assignment problem. Journal of Economic Theory, 100(2):295–328.
- Cho, I. and Kreps, D. (1987). Signaling games and stable equilibria. *The Quarterly Journal* of *Economics*, pages 179–221.
- Coles, P., Kushnir, A., and Niederle, M. (2009). Signaling in matching markets. Working paper, Harvard Business School, Penn State University and Stanford University.
- Crawford, V. and Sobel, J. (1982). Strategic information transmission. *Econometrica*, 50(6):1431–1451.
- d'Aspremont, C. and Peleg, B. (1988). Ordinal Bayesian incentive compatible representations of committees. *Social Choice and Welfare*, 5(4):261–279.
- Farrell, J. and Gibbons, R. (1989). Cheap talk can matter in bargaining. Journal of Economic Theory, 48(1):221–237.
- Fudenberg, D. and Tirole, J. (1991). *Game Theory*. MIT Press.
- Gale, D. and Shapley, L. (1962). College admissions and the stability of marriage. American Mathematical Monthly, 69(1):9–15.
- Lee, R. and Schwarz, M. (2007). Signaling preferences in interviewing markets. Dissertation paper, Harvard Business School and Yahoo! Research.
- Lee, S., Niederle, M., Kim, H.-R., and Kim, W.-K. (2009). Do roses speak louder than words? signaling in internet dating markets. Working paper, University of Maryland, Stanford University, and Korea Marriage Culture Institute.

- Roth, A. (2008). What have we learned from market design? *The Economic Journal*, 118(527):285–310.
- Roth, A. and Xing, X. (1997). Turnaround time and bottlenecks in market clearing: Decentralized matching in the market for clinical psychologists. *Journal of Political Economy*, 105(2):284–329.
- Spence, M. (1973). Job market signaling. Quarterly Journal of Economics, 87(3):355–374.

NOTE DI LAVORO DELLA FONDAZIONE ENI ENRICO MATTEI

Fondazione Eni Enrico Mattei Working Paper Series

Our Note di Lavoro are available on the Internet at the following addresses:

http://www.feem.it/getpage.aspx?id=73&sez=Publications&padre=20&tab=1 http://papers.ssrn.com/sol3/JELJOUR_Results.cfm?form_name=journalbrowse&journal_id=266659

http://ideas.repec.org/s/fem/femwpa.html http://www.econis.eu/LNG=EN/FAM?PPN=505954494

http://ageconsearch.umn.edu/handle/35978

http://www.bepress.com/feem/

NOTE DI LAVORO PUBLISHED IN 2010

GC	1.2010	Cristina Cattaneo: <u>Migrants' International Transfers and Educational Expenditure: Empirical Evidence from</u>
		Albania
SD	2.2010	Fabio Antoniou, Panos Hatzipanayotou and Phoebe Koundouri: <u>Tradable Permits vs Ecological Dumping</u>
SD	3.2010	Fabio Antoniou, Panos Hatzipanayotou and Phoebe Koundouri: <u>Second Best Environmental Policies under</u>
SD	4 2010	Carlo Carraro, Enrica De Cian and Lea Nicita: Modeling Biased Technical Change Implications for Climate
50	4.2010	Policy
IM	5.2010	Luca Di Corato: Profit Sharing under the threat of Nationalization
SD	6.2010	Masako Ikefuji, Jun-ichi Itaya and Makoto Okamura: Optimal Emission Tax with Endogenous Location
		Choice of Duopolistic Firms
SD	7.2010	Michela Catenacci and Carlo Giupponi: Potentials and Limits of Bayesian Networks to Deal with
		Uncertainty in the Assessment of Climate Change Adaptation Policies
GC	8.2010	Paul Sarfo-Mensah and William Oduro: Changes in Beliefs and Perceptions about the Natural Environment
		<u>in the Forest-Savanna Transitional Zone of Ghana: The Influence of Religion</u>
IM	9.2010	Andrea Boitani, Marcella Nicolini and Carlo Scarpa: Do Competition and Ownership Matter? Evidence
		from Local Public Transport in Europe
SD	10.2010	Helen Ding and Paulo A.L.D. Nunes and Sonja Teelucksingh: <u>European Forests and Carbon Sequestration</u>
		Services : An Economic Assessment of Climate Change Impacts
GC	11.2010	Enrico Bertacchini, Walter Santagata and Giovanni Signorello: Loving Cultural Heritage Private Individual
	10.0010	Giving and Prosocial Behavior
SD	12.2010	Antoine Dechezleprêtre, Matthieu Glachant and Yann Ménière: <u>What Drives the International Transfer of</u>
C D	12 2010	Climate Change Mitigation Technologies? Empirical Evidence from Patent Data
SD	13.2010	Andrea Bastianin, Alice Favero and Emanuele Massetti: <u>Investments and Financial Flows Induced by</u>
CD	14 2010	Climate Mitigation Policies
SD	14.2010	Rever Genagn: <u>100 Much Oli</u>
	16.2010	Chiara Fumagaili and Massimo Motta: <u>A Simple Theory of Predation</u>
uc	10.2010	Rinaldo Drau, Adriana Di Liberto and Francesco Pigliaru: <u>Tourism and Development: A Recent</u>
SD	17 2010	<u>Prenomenon Built on Oid (Institutional) Roots:</u>
30	17.2010	MOSE Barriers on Venice Port Activities
SD	18 2010	ZhongXiang Zhang: Climate Change Meets Trade in Promoting Green Growth: Potential Conflicts and
50	10.2010	Svnergies
SD	19.2010	Elisa Lanzi and Ian Sue Wing: Capital Malleability and the Macroeconomic Costs of Climate Policy
IM	20.2010	Alberto Petrucci: Second-Best Optimal Taxation of Oil and Capital in a Small Open Economy
SD	21.2010	Enrica De Cian and Alice Favero: Fairness, Credibility and Effectiveness in the Copenhagen Accord: An
		Economic Assessment
SD	22.2010	Francesco Bosello: Adaptation, Mitigation and "Green" R&D to Combat Global Climate Change. Insights
		From an Empirical Integrated Assessment Exercise
IM	23.2010	Jean Tirole and Roland Bénabou: Individual and Corporate Social Responsibility
IM	24.2010	Cesare Dosi and Michele Moretto: Licences, "Use or Lose" Provisions and the Time of Investment
GC	25.2010	Andrés Rodríguez-Pose and Vassilis Tselios (lxxxvi): <u>Returns to Migration, Education, and Externalities in</u>
		the European Union
GC	26.2010	Klaus Desmet and Esteban Rossi-Hansberg (lxxxvi): <u>Spatial Development</u>
SD	27.2010	Massimiliano Mazzanti, Anna Montini and Francesco Nicolli: Waste Generation and Landfill Diversion
		Dynamics: Decentralised Management and Spatial Effects
SD	28.2010	Lucia Ceccato, Valentina Giannini and Carlo Gipponi: <u>A Participatory Approach to Assess the Effectiveness</u>
		of Responses to Cope with Flood Risk
SD	29.2010	Valentina Bosetti and David G. Victor: <u>Politics and Economics of Second-Best Regulation of Greenhouse</u>
		Gases: The Importance of Regulatory Credibility
IM	30.2010	Francesca Cornelli, Zbigniew Kominek and Alexander Ljungqvist: <u>Monitoring Managers: Does it Matter?</u>
GC	31.2010	Francesco D'Amuri and Juri Marcucci: "Google it!" Forecasting the US Unemployment Rate with a Google
50	22 2010	JOD Search Index
20	32.2010	Francesco Doseilo, Cario Carraro and Enrica De Cian: <u>Climate Policy and the Optimal Balance between</u>
		<u>willigation, Adaptation and Unavoided Damage</u>

SD	33.2010	Enrica De Cian and Massimo Tavoni: <u>The Role of International Carbon Offsets in a Second-best Climate</u>
SD	34.2010	ZhongXiang Zhang: The U.S. Proposed Carbon Tariffs, WTO Scrutiny and China's Responses
IM	35.2010	Vincenzo Denicolò and Piercarlo Zanchettin: Leadership Cycles
SD	36.2010	Stéphanie Monjon and Philippe Quirion: <u>How to Design a Border Adjustment for the European Union</u> Emissions Trading System?
SD	37.2010	Meriem Hamdi-Cherif, Céline Guivarch and Philippe Quirion: <u>Sectoral Targets for Developing Countries:</u> Combining "Common but Differentiated Responsibilities" with "Meaningful participation"
IM	38.2010	G. Andrew Karolyi and Rose C. Liao: <u>What is Different about Government-Controlled Acquirers in Cross</u> - Border Acquisitions?
GC	39,2010	Kietil Biorvatn and Alireza Naghavi: Rent Seekers in Rentier States: When Greed Brings Peace
GC	40.2010	Andrea Mantovani and Alireza Naghavi: Parallel Imports and Innovation in an Emerging Economy
SD	41.2010	Luke Brander, Andrea Ghermandi, Onno Kuik, Anil Markandya, Paulo A.L.D. Nunes, Marije Schaafsma and Alfred Wagtendonk: <u>Scaling up Ecosystem Services Values: Methodology, Applicability and a Case</u>
SD	42.2010	Valentina Bosetti, Carlo Carraro, Romain Duval and Massimo Tavoni: <u>What Should We Expect from</u> <u>Innovation? A Model-Based Assessment of the Environmental and Mitigation Cost Implications of Climate-</u>
		Related R&D
SD	43.2010	Frank Vöhringer, Alain Haurie, Dabo Guan, Maryse Labriet, Richard Loulou, Valentina Bosetti, Pryadarshi R. Shukla and Philippe Thalmann: <u>Reinforcing the EU Dialogue with Developing Countries on Climate</u>
GC	44.2010	Change Mitigation Angelo Antoci, Pier Luigi Sacco and Mauro Sodini: <u>Public Security vs. Private Self-Protection: Optimal</u>
IM	45 2010	<u>I axation and the Social Dynamics of Fear</u> Luca Enriques: European Takeover Law, The Case for a Neutral Approach
SD	46.2010	Maureen L. Cropper, Yi Jiang, Anna Alberini and Patrick Baur: Getting Cars Off the Road: The Cost-
02		Effectiveness of an Episodic Pollution Control Program
IM	47.2010	Thomas Hellman and Enrico Perotti: The Circulation of Ideas in Firms and Markets
IM	48.2010	James Dow and Enrico Perotti: <u>Resistance to Change</u>
SD	49.2010	Jaromir Kovarik, Friederike Mengel and José Gabriel Romero: (Anti-) Coordination in Networks
SD	50.2010	Helen Ding, Silvia Silvestri, Aline Chiabai and Paulo A.L.D. Nunes: <u>A Hybrid Approach to the Valuation of</u> <u>Climate Change Effects on Ecosystem Services: Evidence from the European Forests</u>
GC	51.2010	Pauline Grosjean (lxxxvii): <u>A History of Violence: Testing the 'Culture of Honor' in the US South</u>
GC	52.2010	Paolo Buonanno and Matteo M. Galizzi (Ixxxvii): <u>Advocatus, et non latro? Testing the Supplier-Induced-Demand Hypothesis for Italian Courts of Justice</u>
GC	53.2010	Gilat Levy and Ronny Razin (Ixxxvii): <u>Religious Organizations</u>
GC	54.2010	Matteo Cervellati and Paolo Vanin (lxxxvii): <u>"Thou shalt not covet": Prohibitions, Temptation and Moral Values</u>
GC	55.2010	Sebastian Galiani, Martín A. Rossi and Ernesto Schargrodsky (lxxxvii): <u>Conscription and Crime: Evidence</u> from the Argentine Draft Lottery
GC	56.2010	Alberto Alesina, Yann Algan, Pierre Cahuc and Paola Giuliano (Ixxxvii): <u>Family Values and the Regulation of</u> <u>Labor</u>
GC	57.2010	Raquel Fernández (lxxxvii): <u>Women's Rights and Development</u>
GC	58.2010	Tommaso Nannicini, Andrea Stella, Guido Tabellini, Ugo Troiano (Ixxxvii): <u>Social Capital and Political</u> <u>Accountability</u>
GC	59.2010	Eleonora Patacchini and Yves Zenou (Ixxxvii): Juvenile Delinquency and Conformism
GC	60.2010	Gani Aldashev, Imane Chaara, Jean-Philippe Platteau and Zaki Wahhaj (Ixxxvii): <u>Using the Law to Change the Custom</u>
GC	61.2010	Jeffrey Butler, Paola Giuliano and Luigi Guiso (Ixxxvii): <u>The Right Amount of Trust</u>
SD	62.2010	Valentina Bosetti, Carlo Carraio and Massimo Tavoni: <u>Alternative Paths toward a Low Carbon World</u>
SD	63.2010	Kelly C. de Bruin, Rob B. Dellink and Richard S.J. Tol: <u>International Cooperation on Climate Change</u> <u>Adaptation from an Economic Perspective</u>
IM	64.2010	Andrea Bigano, Ramon Arigoni Ortiz, Anil Markandya, Emanuela Menichetti and Roberta Pierfederici: <u>The</u> <u>Linkages between Energy Efficiency and Security of Energy Supply in Europe</u>
SD	65.2010	Anil Markandya and Wan-Jung Chou: <u>Eastern Europe and the former Soviet Union since the fall of the</u> Berlin Wall: <u>Review of the Changes in the Environment and Natural Resources</u>
SD	66.2010	Anna Alberini and Milan Ščasný: <u>Context and the VSL: Evidence from a Stated Preference Study in Italy and the Czech Republic</u>
SD	67.2010	Francesco Bosello, Ramiro Parrado and Renato Rosa: <u>The Economic and Environmental Effects of an EU</u> Ban on Illegal Logging Imports. Insights from a CGE Assessment
IM	68.2010	Alessandro Fedele, Paolo M. Panteghini and Sergio Vergalli: <u>Optimal Investment and Financial Strategies</u> under Tax Rate Uncertainty
IM	69.2010	Carlo Cambini, Laura Rondi: <u>Regulatory Independence and Political Interference</u> : <u>Evidence from EU Mixed-</u> <u>Ownership Utilities' Investment and Debt</u>
SD	70.2010	Xavier Pautrel: Environmental Policy, Education and Growth with Finite Lifetime: the Role of Abatement Technology
SD	71.2010	Antoine Leblois and Philippe Quirion: <u>Agricultural Insurances Based on Meteorological Indices:</u> <u>Realizations, Methods and Research Agenda</u>
IM IM	72.2010 73.2010	Bin Dong and Benno Torgler: <u>The Causes of Corruption: Evidence from China</u> Bin Dong and Benno Torgler: <u>The Consequences of Corruption: Evidence from China</u>

IM	74 2010	Fereveloup Verdineiad and Vasaman Gorij: The Oil-Based Economies International Research Project. The
IIVI	74.2010	Case of Iran
CC	75 2010	<u>Case of Irall.</u>
uc	75.2010	Scenes Michaelopoulos, Amitza Ragnavi and Giovanni Prarolo (IXXXII). <u>Trade and Geography III die</u>
SD.	76 2010	Zhone Vienz Zhenzi China in the Transition to a Low Carbon Economy
30	70.2010	Zhong Zhang: <u>Chima in the transition to a Low-Carbon Economy</u>
SD	//.2010	Valentina latolla, Massimiliano Mazzanti and Francesco Nicolli: <u>Are You SURE You Want to Waste Policy</u>
		Chances? Waste Generation, Landfill Diversion and Environmental Policy Effectiveness in the EU15
IM	78.2010	Jean Tirole: <u>Illiquidity and all its Friends</u>
SD	79.2010	Michael Finus and Pedro Pintassilgo: International Environmental Agreements under Uncertainty: Does
		the Veil of Uncertainty Help?
SD	80.2010	Robert W. Hahn and Robert N. Stavins: The Effect of Allowance Allocations on Cap-and-Trade System
		Performance
SD	81 2010	Francisco Alpizar Fredrik Carlsson and Maria Naranio (Ixxxviji): The Effect of Risk Ambiguity and
50	01.2010	Coordination on Farmers' Adaptation to Climate Change: A Framed Field Experiment
۲D	82 2010	Coordination of markets Adaptation to cancer angle. A market field Experiment
30	82.2010	Shardu Agrawaa ahu Maelis Carraro (ixxxviii). Assessing the Role of Micromatice in rostering Adaptation
CD	02 2010	to chimate Change
SD	83.2010	Wolfgang Lutz (IxxxvIII): Improving Education as Key to Enhancing Adaptive Capacity in Developing
		Countries
SD	84.2010	Rasmus Heltberg, Habiba Gitay and Radhika Prabhu (lxxxviii): <u>Community-based Adaptation: Lessons</u>
		from the Development Marketplace 2009 on Adaptation to Climate Change
SD	85.2010	Anna Alberini, Christoph M. Rheinberger, Andrea Leiter, Charles A. McCormick and Andrew Mizrahi:
		What is the Value of Hazardous Weather Forecasts? Evidence from a Survey of Backcountry Skiers
SD	86.2010	Anna Alberini, Milan Ščasný, Dennis Guignet and Stefania Tonin: The Benefits of Contaminated Site
02	00.2010	Cleanup Revisited: The Case of Naples and Caserta Italy
CC	97 2010	David Sorte Mansah William Odwar Erediak Antah Eredua and Stanhan Amisah. Traditional
uc	87.2010	Paul Sario-Mensan, William Oduro, Fredrick Anton Fredua and Stephen Antisan. Traditional
		Representations of the Natural Environment and Biodiversity Conservation: Sacred Groves in Ghana
IM	88.2010	Gian Luca Clementi, Thomas Cooley and Sonia Di Giannatale: <u>A Theory of Firm Decline</u>
IM	89.2010	Gian Luca Clementi and Thomas Cooley: <u>Executive Compensation: Facts</u>
GC	90.2010	Fabio Sabatini: Job Instability and Family Planning: Insights from the Italian Puzzle
SD	91.2010	ZhongXiang Zhang: Copenhagen and Beyond: Reflections on China's Stance and Responses
SD	92.2010	ZhongXiang Zhang: Assessing China's Energy Conservation and Carbon Intensity: How Will the Future
		Differ from the Past?
SD	93,2010	Daron Acemoglu, Philippe Aghion, Leonardo Bursztyn and David Hemous: The Environment and Directed
02	2012010	Technical Change
SD	94 2010	Valaria Costantiai and Massimiliano Mazzanti. On the Green Side of Trade Competitiveness?
30	94.2010	Valeria Costantini and massiminatio mazzanti. On the Green side of trade competitiveness:
15.4	05 2010	Environmental Policies and innovation in the EU
IM	95.2010	Vittoria Cerasi, Barbara Chizzolini and Marc Ivaldi: <u>The Impact of Mergers on the Degree of Competition</u>
		in the Banking Industry
SD	96.2010	Emanuele Massetti and Lea Nicita: The Optimal Climate Policy Portfolio when Knowledge Spills Across
		Sectors
SD	97.2010	Sheila M. Olmstead and Robert N. Stavins: Three Key Elements of Post-2012 International Climate Policy
		Architecture
SD	98.2010	Lawrence H. Goulder and Robert N. Stavins: Interactions between State and Federal Climate Change
		Policies
IM	99 2010	Philippe Aghion John Van Reenen and Luigi Zingales: Innovation and Institutional Ownership
<u> </u>	100 2010	Angele Antoni Eshio Selatini and Auge Sedini. The Selavia Sundrama Seciel Control in a Crawing
uc	100.2010	Angelo Antoci, Fabio Sabatim and Mauro Sodim: <u>The Solaria Syndrome: Social Capital in a Growing</u>
60	101 0010	<u>Hyper-tecnnological Economy</u>
SD	101.2010	Georgios Kossioris, Michael Plexousakis, Anastasios Xepapadeas and Aart de Zeeuw: On the Optimal
		Taxation of Common-Pool Resources
SD	102.2010	ZhongXiang Zhang: Liberalizing Climate-Friendly Goods and Technologies in the WTO: Product Coverage,
		Modalities, Challenges and the Way Forward
SD	103.2010	Gérard Mondello: Risky Activities and Strict Liability Rules: Delegating Safety
GC	104.2010	João Ramos and Benno Torgler: Are Academics Messy? Testing the Broken Windows Theory with a Field
		Experiment in the Work Environment
INA	105 2010	Mauria Cinechai Erangesca Severini Claudio Socci and Posita Pretaroli: The Economic Impact of the
IIVI	103.2010	Construction Made through the Mars Multiplice Assessed
CD	100 2010	Green Certificate Market through the Macro Multiplier Approach
SD	106.2010	Joele Noally: improving the Energy-Efficiency of Buildings: The Impact of Environmental Policy on
		lechnological Innovation
SD	107.2010	Francesca Sanna-Randaccio and Roberta Sestini: <u>The Impact of Unilateral Climate Policy with Endogenous</u>
		Plant Location and Market Size Asymmetry
SD	108.2010	Valeria Costantini, Massimiliano Mozzanti and Anna Montini: Environmental Performance and Regional
		Innovation Spillovers
IM	109.2010	Elena Costantino, Maria Paola Marchello and Cecilia Mezzano: Social Responsibility as a Driver for Local
		Sustainable Development
GC	110 2010	Marco Percoco: Path Dependence Institutions and the Density of Economic Activities: Evidence from
30	110.2010	Italian Citize
SD.	111 0010	Italian Critics
20	111.2010	Sonja S. reelucksingn and Paulo A.L.D. Nunes: <u>Biodiversity valuation in Developing Countries: A Focus</u>
CD	110 0010	on small Island Developing States (SIDS)
50	112.2010	Znongxiang Znang: In What Format and under What Timetrame Would China Take on Climate
		Commitments? A Roadmap to 2050

- SD 113.2010 Emanuele Massetti and Fabio Sferra: <u>A Numerical Analysis of Optimal Extraction and Trade of Oil under</u> <u>Climate Policy</u>
- IM
 114.2010 Nicola Gennaioli, Andrei Shleifer and Robert Vishny: <u>A Numerical Analysis of Optimal Extraction and Trade of Oil under Climate Policy</u>
- GC 115.2010 Romano Piras: <u>Internal Migration Across Italian regions: Macroeconomic Determinants and</u> <u>Accommodating Potential for a Dualistic Economy</u>
- SD 116.2010 Messan Agbaglah and Lars Ehlers (lxxxiv): Overlapping Coalitions, Bargaining and Networks
- SD 117.2010 Pascal Billand, Christophe Bravard, Subhadip Chakrabarti and Sudipta Sarangi (lxxxiv):<u>Spying in Multi-</u> market Oligopolies
- SD 118.2010 Roman Chuhay (lxxxiv): <u>Marketing via Friends: Strategic Diffusion of Information in Social Networks with</u> <u>Homophily</u>
- SD 119.2010 Françoise Forges and Ram Orzach (lxxxiv): <u>Core-stable Rings in Second Price Auctions with Common</u> <u>Values</u>
- SD 120.2010 Markus Kinateder (lxxxiv): The Repeated Prisoner's Dilemma in a Network
- SD 121.2010 Alexey Kushnir (lxxxiv): <u>Harmful Signaling in Matching Markets</u>

(lxxxvi) This paper was presented at the Conference on "Urban and Regional Economics" organised by the Centre for Economic Policy Research (CEPR) and FEEM, held in Milan on 12-13 October 2009.

(lxxxvii) This paper was presented at the Conference on "Economics of Culture, Institutions and Crime" organised by SUS.DIV, FEEM, University of Padua and CEPR, held in Milan on 20-22 January 2010.

(Ixxxviii) This paper was presented at the International Workshop on "The Social Dimension of Adaptation to Climate Change", jointly organized by the International Center for Climate Governance, Centro Euro-Mediterraneo per i Cambiamenti Climatici and Fondazione Eni Enrico Mattei, held in Venice, 18-19 February 2010.

(lxxxiv) This paper was presented at the 15th Coalition Theory Network Workshop organised by the Groupement de Recherche en Economie Quantitative d'Aix-Marseille, (GREQAM), held in Marseille, France, on June 17-18, 2010.